Author's response to reviews

Title: Unravelling Developmental Disregard in children with unilateral Cerebral Palsy by measuring Event Related Potentials during a simple and complex task

Authors:

Ingar M Zielinski (i.zielinski@pwo.ru.nl)
Marijtje LA Jongsma (m.jongsma@donders.ru.nl)
Marjolein Baas (cm.baas@pwo.ru.nl)
Pauline BM Aarts (p.aarts@maartenskliniek.nl)
Bert Steenbergen (b.steenbergen@pwo.ru.nl)

Version: 2 Date: 26 September 2013

Author's response to reviews:

Ingar M Zielinski
Radboud University Nijmegen
Behavioural Science Institute, PO Box 9104
6500 HE Nijmegen, the Netherlands
tel.: +31-24-3615812
e-mail: i.zielinski@pwo.ru.nl

Josefino M. Rodis
Journal Editorial Office
BioMed Central

September 23, 2013

Dear Mr. Rodis,

Thank you for your email dated September 24th. Based on the instructions provided in your email, we uploaded the file of the revised manuscript on the journal's website.

As you can see, based on your comment we have revised the manuscript by modifying the consent statement of the Participant section as well as the first sentence of the Procedure section and the second sentence of the Participant section.

By applying the modification concerning the consent statement we wanted to clarify that all parents of the participants gave consent for using individual participant information. Furthermore, by modifying the information concerning the procedure and the second sentence of the Participant section, we wanted to point out that the individual information as seen in table 1 were all acquired
during the experimental setting. We therefore want to clarify that we did not use any details of the participants that were not obtained during our own data acquisition.

Sincerely,
Ingar M. Zielinski, MSc